Author's response to reviews

Title: Effectiveness of community based safe motherhood promoters in improving the utilization of obstetric care. The case of Mtwara Rural District in Tanzania

Authors:

Declare Mushi (declbety@yahoo.com)
Rose Mpembeni (rmpembeni@muhas.ac.tz)
Albrecht Jahn (albrecht.jahn@urz.uni-heidelberg.de)

Version: 10 Date: 29 March 2010

Author's response to reviews: see over
Dear BMC Series Editorial Production Team

Refer to our manuscript Titled: Effectiveness of community based safe motherhood promoters in improving the utilization of obstetric care. The case of Mtwara Rural District in Tanzania
Journal: BMC Pregnancy and Childbirth
MS ID: 6304547829746396

Thanks very much for your e-mail and the feedback. We have done all the necessary editing for final publication. We have also put the tables in separate pages after the references.

Thanks for your cooperation.

Sincerely yours

Declare Mushi